Author's response to reviews

Title: The first report of severe anemia associated with acute renal failure and adult minimal change disease: A case report

Authors:

Yimei YQ Qian Dr. (yqian2005@hotmail.com)
Sushil SK Mehandru Dr. (SKmehandru@meridianhealth.com)
Nancy NG Gornish Dr. (NGornish@meridianhealth.com)
Elliot EF Frank Dr. (EFrank@meridianhealth.com)

Version: 2 Date: 30 May 2008

Author's response to reviews:

1. included email id’s of all the authors in the title page
2. removed the running title and word count sections from the title page
3. removed the Support and Financial Disclosure Declaration from the title page
4. included the information on patient's ethnic origin in the case presentation section of the abstract
5. removed the key words section a from the abstract
6. removed the abbreviations from the abstract
7. included the information on patient's ethnic origin in the case presentation section
8. included a Conclusions section
9. included a competing interests section.
10. included an Authors’ Contributions section at the end of the manuscript, before the reference list.
11. included a Consent section at the end of the manuscript, before the reference list.
12. rephrase "Legend" as "Figure Legend"